Author's response to reviews

Title: Benzyl butyl phthalate induces migration, invasion, and angiogenesis of Huh7 hepatocellular carcinoma cells through nongenomic AhR/G-protein signaling

Authors:

Cheng-Fang Tsai (chenfung2001@yahoo.com.tw)
Tsung-Hua Hsieh (pelagice@yahoo.com.tw)
Jau-Nan Lee (jaunanlee@hotmail.com)
Chia-Yi Hsu (husonweihsu@hotmail.com)
Yu-Chih Wang (joelovefu@gmail.com)
Feng-Jie Lai (D950819@mail.chimei.org.tw)
Kung-Kai Kuo (kuoksfo@yahoo.com.tw)
Hua-Lin Wu (halnwu@mail.ncku.edu.tw)
Eing-Mei Tsai (tsaieing@yahoo.com)
Po-Lin Kuo (kuopolin@seed.net)

Version: 2 Date: 12 November 2013

Author's response to reviews: see over
Dear Editors:

Please consider the enclosed manuscript entitled “Benzyl butyl phthalate induces migration, invasion, and angiogenesis of Huh-7 hepatocellular carcinoma cells through nongenomic AhR/G protein signaling” for publication in your journal. This study demonstrates that phthalate exposure results in an increase in tumor progression of hepatoma through nongenomic AhR action and G-protein signaling. These data provide valuable information regarding the risk of phthalate exposure to human health and suggest therapeutic targets.

All authors have read the manuscript, agree that the work is ready for submission to a journal, and accept responsibility for the manuscript’s contents. The content of the manuscript is original, and it has not been published or accepted for publication, either in whole or in part, in any form. No part of the manuscript is currently under consideration for publication elsewhere.

Conflict of interest statement:
The authors declare that there are no conflicts of interest.

Thank you for your attention and consideration.

Sincerely yours,
Eing-Mei Tsai